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Highlights:

We hear from our correspondent-attendees at the International FEL Conference, held this week in Trieste, that the seven papers presented by the JLab FEL team were well received. (The titles of these presentations were given in last week’s report).

On the home front in Newport News, the FEL shut-down activities are proceeding well.

Highlights of this week’s accomplishments:

- removal of the optical klystron wiggler and temporary removal of the incoming/ongoing telescopes in preparation for the installation of the new EM wiggler optimized for short wavelength (1-3 micron) operation

- replacement of the old style poly rf windows with the new ceramic versions on Cryomodule 1

 (these posed limits to our high current operation in the last run)

- all components for the first phase of the optical transport system have arrived

- the design and layout for the THz suppression chicane has been completed.

We wish our colleagues and readers a very pleasant Labor Day weekend. 

Management:

This week (Aug. 30-31), the management of the Lab was reviewed during our biannual Institutional Management Review sponsored by DOE and SURA. The review was chaired by Dr. Charles Shank, the recently retired director of Lawrence Berkeley National Laboratory.

The FEL program was commended for the strength of the collaboration with DoD and its execution and accomplishments during the past year.

Cost books and staffing plans were prepared for the FY05 UV Upgrade effort and the proposed FY05-07 effort that was submitted to ONR last month for their review. 

Planning/progress meetings were held this week on the UV effort and the current shut-down activities.  The punch list for the shut-down contains 120 items that are being tracked for completion during this period. We are making good progress on the high priority major activities that comprise the critical path of the shut-down:

- removal (complete) and replacement of the wiggler

- replacement of the warm windows on CM1 (complete this week)

- design (complete), fab and installation of the THz suppression chicane

- installation of the first phase of the optical transport system (all parts are on site)

WBS 4 (Injector):

This week we have looked at the vacuum conditions in the gun. Our extractor gauge corrected for hydrogen indicates a pressure of 6.5X10-11 torr. The RGA shows the residual gas to be 99% hydrogen.

WBS 5/6 (SRF/RF):

Zone 1 - The leak testing and purging of the Quarter unit has started. 

Zone 2 - The poly warm windows have been replaced with ceramic ones.

The waveguide and cables have been reconnected.

Zone 3 - I replaced a leaky hose in the HPA.

Zone 4 - No known problems.

All RF was operated to the Filament state, except for the Buncher.  The Buncher LCW is apart for the SRF work on the Quarter.  The PSS and cold cavities are needed to take the RF systems to the HV and RFON states during our hot check prior to restart of operations.

WBS 8 (Instrumentation):

Much time was spent in the vault supporting the Installation and Vacuum groups with preparations for the new wiggler installation. Disassembly of beam line components continued and the general removal of unnecessary items in the vault. 

ANALOG MONITORING SYSTEM (AMS)/Video - More of the crosspoint input BNC 
boards have been ordered. These are the passive PCBs that convert the coaxial system inputs and outputs over to the euro-connectors that the crosspoint system uses internally. A new product has been identified for performing text overlay for the video system. We will be using in-system-programmable text overlay video modules to each of the video outputs which will allow the re-labeling of the output as required using EPICS. 

A flow interlock for the Terahertz Vacuum module in the vault was installed to prevent over heating of the lamps due to insufficient water flow.  Final calibration to ensure proper flow level is in progress. Additional BLM signal and power cables were installed for the 0F region.  
Full commissioning of the  LPSS master is nearly complete. The vendor drawings for the User Lab Interface PLC's are near completion  and should be submitted to the FEL next week for review. The card reader is now communicating perfectly with the software tests that were created. All commands tested are working, along with the proper retrieval of the response codes.  Response codes of varying lengths are handled correctly to prevent all communication errors. Documentation of the system is progressing as changes are being made. 

Controls Database (Devlore) - A lot of great programming completed by our students over the summer has not yet been made available in the current version of the DevLore Controls database system. The upgrade will make the whole system much easier to navigate between system components. We have also determined the right balance of security and accessibility in the system. We are moving many of our raw work files and folders to on-line, we cannot continue to have open anonymous 'read-only' access to all areas of the site. The solution being implemented will make designated folders only accessible to authenticated JLAB users (even if they are not on-site). 

WBS 9 (Beam Transport):
Wiggler

Electromagnetic Wiggler for 2.8µ

•
The test stand software was upgraded during the week so tests were unable to be run.

•
At the FEL Conference this week, two institutions expressed interest in extending the water jet cut coil pack technology to either making their wiggler or recoiling an existing wiggler.

     We will work out this option of technology transfer with the DOE and ONR program offices.

Installing the EM Wiggler and Decompression Chicane to Reduce Downstream THz

•   
Layout of positions for the new beam line elements continued.

Sextupole (SF)

•     We continue putting the procurement package together

UV Branch Line

•
We started ordering the replacement GW Dipoles for the UV by-pass.

WBS 11 (Optics): 

Progress occurred in the following areas:

Cryomirror:

We reinstalled the optic with a new heater to simulate the loading by the FEL.  Various experimental problems, including the loss of a good retroreflection, and a vacuum leak precluded obtaining more data.  With

the loss of beamline diagnostics for awhile for installation of the new wiggler, we will concentrate on measuring flow rates, and plan for the next round of experiments.

2.8 and 1.06 micron mirrors:

Silicon substrates will be shipped over the next week, with sapphire optics shipping the week afterward.  We should still be able to make the installation window for the 2.8 um mirrors.  Results from calorimetry experiments on 

1.06 um coatings will determine whether we can still install 1.06 um mirrors in September.

Scraper outcoupler:

Our results and interpretation were presented at the International FEL conference.  The high outcoupling efficiency, relative to the hole-outcoupler mirrors, interested a number of participants, and was highlighted in an overview talk by M. Poole as one of the important new concepts presented in this conference.

Other activities:

Survey and Alignment laid out the alignment points on the FEL vault ceiling in preparation for installation of the OTS hardware.  We had an intense round of design for stray light shields and improved water connections to the optical cavity optics.  We will get those into the shop so we can install them this month.  The cryopump assembly is coming along and will be tested this month as well.  The installation of the transport for the streak camera also progress.  All the hardware for the first phase of the OTS system is now in house.

Terahertz:

During this period the M2, M3 and M4 mirrors are being re-polished. It has been decided to add a spring to the kinematic mounts for M2, M3 and M4.  This will provide an additional seating force for the ball, and make the horizontally mounted mirrors more secure.  Components for the second safety shutter arrived and are being assembled.
